Author's response to reviews

Title: SNPs and real-time quantitative PCR method for constitutional allelic copy number determination, the VPREB1 marker case

Authors:

Marcello Frigerio (marcello.frigerio@gmail.com)
Elena Passeri (thesparrows@tiscali.it)
Tiziana de Filippis (defitiz@yahoo.it)
Daniela Rusconi (d.rusconi@auxologico.it)
Rea Valaperta (rea.valaperta@unimi.it)
Mario Carminati (carminati.mario@lycos.com)
Anita Donnangelo (nitina@libero.it)
Elena Costa (elena.costa@grupposandonato.it)
Luca Persani (luca.persani@unimi.it)
Palma Finelli (palma.finelli@unimi.it)
Sabrina Corbetta (sabrina.corbetta@unimi.it)

Version: 6 Date: 22 April 2011

Author's response to reviews: see over
To Editor-in-Chief
BMC Medical Genetics

San Donato Milanese, April 22, 2011

We realized a problem in the revised version submitted for the proofing stage of the manuscript entitled “SNPs and real-time quantitative PCR method for constitutional allelic copy number determination, the VPREB1 qPCR marker case” by Frigerio M et al, to BMC Medical Genetics as Original Article.

In the manuscript, legend to figure 1 has been changed with legend to figure 3 and viceversa. Any further change has been provided throughout the manuscript.

Looking forward to hearing from you soon

Sabrina Corbetta, MD, PhD
Endocrinology and Diabetology Unit
Dept.Medical-Surgical Sciences
University of Milan
IRCCS Policlinico San Donato
Piazza Malan 2
20097 San Donato Milanese (MI)
Italy
Phone/Fax +39 02 52774444
E-mail sabrina.corbetta@unimi.it